**Author's response to reviews**

**Title:** Procalcitonin guided antibiotic therapy and hospitalization in patients with lower respiratory tract infections: a prospective, multicenter, randomized controlled trial

**Authors:**
- Philipp Schuetz (schuetzp@uhbs.ch)
- Mirjam Christ-Crain (christmj@bluewin.ch)
- Marcel Wolbers (MWolbers@uhbs.ch)
- Ursula Schild (Schildu@uhbs.ch)
- Robert Thomann (thomannr@uhbs.ch)
- Claudine Falconnier (falconnierc@uhbs.ch)
- Isabelle Widmer (isabelle.widmer@ksl.ch)
- Stephanie Neidert (stefanie.neidert@ksl.ch)
- Claudine A Blum (claudine.blum@ksa.ch)
- Ronald Schoenenberger (rschoenenberger_so@spital.ktso.ch)
- Christoph Henzen (christoph.henzen@ksl.ch)
- Thomas Bregenzer (thomas.bregenzer@ksa.ch)
- Claus Hoess (claus.hoess@stgag.ch)
- Martin Krause (martin.krause@stgag.ch)
- Heiner C Bucher (Bucherr@uhbs.ch)
- Werner Zimmerli (werner.zimmerli@ksli.ch)
- Beat Mueller (happy.mueller@unibas.ch)

**Version:** 2  **Date:** 3 July 2007

**Author's response to reviews:** see over
Changes for the study protocol

Please remove the running title and word count from the title page.
→ ok

The ProHOSP study group information - Please remove this from the title page and move to the acknowledgments section.
→ ok

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names. Can we ask you to link each author to their corresponding affiliation with the use of superscript numbers.
→ ok

Abstract - The trial registration number must be mentioned in the abstract and removed from the title. The last section of the abstract must be titled Trial Registration: listing the trial registry and the unique identifying number. Please note that there must be no space between the letters and numbers of the trial registration number e.g. ISRCTN12345678.
→ ok

Footer - Please remove the footer from the manuscript.
→ ok

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT. Please also remove the number ‘1’ from the abstract heading.
→ ok

Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. Please remember to also update the Abstract details on the submission page.
→ we restructured the abstract accordingly

Website addresses - Websites must be removed from the text, and cited from the main reference list only. If you are using referencing software such as EndNote or Refman, please ensure that your references are entered through the software rather than included in the manuscript reference list manually. Please present your reference in the following structure, e.g. 17. Website title [www.website.com].
→ we restructured the websites in the references accordingly

Contributors - Authors’ contributions.
→ the Authors contributions are mentioned
Funding sources - Please move this to the acknowledgements. 
→ ok

Please move the figures and tables sections to a position below the references. 
→ ok

Table title and legend - The table title should appear above the table. The legend (or key) should appear below the table. Please remove the table files which are uploaded as additional files. 
→ ok

Figures - We note that the figures have been included in the manuscript file. Please only upload the figures as separate figure files using the upload form on the submission system, and delete the figure from the manuscript file. Please leave the figure legends in the manuscript (after the references). 
→ ok